The study has a large sample number for the selected patients which is a positive point about the study. The methodology is adequate. The study only proves the MRSA eradication therapy to be inadequate in completely reducing incidence of PJI. There could be several factors contributing to it, from inadequate dosage, non-compliance of patients, inadequate gap from treatment and arthroplasty etc. However, these aspects are not explored in the study. So only a speculation can be made as to the reasoning for the increased incidence of PJI. I understand this is a retrospective study and so that could not be explored. Considering the above matters, the study achieves its aim of proving the current eradication regime to be inadequate. The study would still merit publication in my opinion as it further justifies the need for more prospective studies on the matter.

Reviewers comments

I have reviewed the manuscript.

Author contributions added.

I have reviewed the references and addressed concerns.

Articles highlights added.

I have attached the required files.